Author's response to reviews

Title: Randomized Pharmacokinetic Evaluation of Different Rifabutin Doses in African HIV-infected Tuberculosis Patients on Lopinavir/Ritonavir-based Antiretroviral Therapy

Authors:

Suhashni Naiker (sunaiker@gmail.com)
Cathy Connolly (Cathy.Connolly@mrc.ac.za)
Lubbe Wiesner (lubbe.wiesner@uct.ac.za)
Tracey Kellerman (tkellermann@mweb.co.za)
Tarylee Reddy (Tarylee.Reddy@mrc.ac.za)
Anthony Harries (adharries@theunion.org)
Helen McIleron (helen.mcilleron@uct.ac.za)
Christian Lienhardt (lienhardtc@who.int)
Alexander S Pym (alex.pym@k-rith.org)

Version: 2
Date: 7 February 2014

Author's response to reviews: see over
07 February 2014

Dear Editor,

Re: Revision to MS: 7837918311199394

(1) Results: Please separate the Results from Methods. The Results should be placed after the Methods.

These are separate with methods starting on page 4 and the results on page 7. Method is now written in upper case to emphasis this distinction

(2) Competing Interests:

A competing interests section has been added to the manuscript

(3) Authors’ Contributions:

An authors’ contribution section has been added to the manuscript

(4) Acknowledgements: ‘

An acknowledgements section has been added to the manuscript

(5) Tables:

Tables have been moved

Yours sincerely,

Alexander Pym, MD, MRCP, PhD
Associate Investigator